Author's response to reviews

Title: Epoetin administrated after cardiac surgery: effects on renal function and inflammation in a randomized controlled study

Authors:

Sophie de Seigneux (sophie.deseigneux@hcuge.ch)
Belen Ponte (belen.ponte@hcuge.ch)
Lucien Weiss (lucien.weiss@hcuge.ch)
Jerome Pugin (jerome.pugin@hcuge.ch)
Jacques-André Romand (romand@etat.ge.ch)
Pierre-Yves Martin (pierre-yves.martin@hcuge.ch)
Patrick Saudan (Patrick.saudan@hcuge.ch)

Version: 2 Date: 18 May 2012

Author's response to reviews:

Dear Dr Henderson,

We provide here a revised version of our manuscript including the following changes:
-We added the consort checklist to the manuscript
-The abstract was revised according to CONSORT
-A statement regarding the calculation of the power of the study was included
-We provided the name of the ethical commission
-We placed the funding source in the acknowledgment section

We hope that you will consider this revised version for publication in BMC Nephrology.

Sincerely yours
Patrick Saudan